RESEARCH ARTICLE

Early Indicators of Fatal Leptospirosis during the 2010 Epidemic in Puerto Rico

Tyler M. Sharp^{1,2}*, Brenda Rivera García³, Janice Pérez-Padilla², Renee L. Galloway⁴, Marta Guerra⁴, Kyle R. Ryff³, Dana Haberling⁴, Sharada Ramakrishnan⁴, Sean Shadomy⁴, Dianna Blau⁵, Kay M. Tomashek², William A. Bower⁴

1 Epidemic Intelligence Service, Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America, 2 Dengue Branch, Division of Vector-Borne Diseases, Centers for Disease Control and Prevention, San Juan, Puerto Rico, 3 Puerto Rico Department of Health, San Juan, Puerto Rico, 4 Bacterial Special Pathogens Branch, Division of High Consequence Pathogens, Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America, 5 Infectious Diseases Pathology Branch, Division of High Consequence Pathogens control and Prevention, Atlanta, Georgia, United States of America, 5 Infectious Disease Seathology Branch, Division of High Consequence Pathogens and Pathology, Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America

* tsharp@cdc.gov

Abstract

Background

Leptospirosis is a potentially fatal bacterial zoonosis that is endemic throughout the tropics and may be misdiagnosed as dengue. Delayed hospital admission of leptospirosis patients is associated with increased mortality.

Methodology/Principal Findings

During a concurrent dengue/leptospirosis epidemic in Puerto Rico in 2010, suspected dengue patients that tested dengue-negative were tested for leptospirosis. Fatal and non-fatal hospitalized leptospirosis patients were matched 1:1–3 by age. Records from all medical visits were evaluated for factors associated with fatal outcome. Among 175 leptospirosis patients identified (4.7 per 100,000 residents), 26 (15%) were fatal. Most patients were older males and had illness onset during the rainy season. Fatal case patients first sought medical care earlier than non-fatal control patients (2.5 vs. 5 days post-illness onset [DPO], p < 0.01), but less frequently first sought care at a hospital (52.4% vs. 92.2%, p < 0.01). Although fatal cases were more often diagnosed with leptospirosis at first medical visit (43.9% vs. 9.6%, p = 0.01), they were admitted to the hospital no earlier than non-fatal controls (4.5 vs. 6 DPO, p = 0.31). Cases less often developed fever (p = 0.03), but more often developed jaundice, edema, leg pain, hemoptysis, and had a seizure (p \leq 0.03). Multivariable analysis of laboratory values from first medical visit associated with fatal outcome included increased white blood cell (WBC) count with increased creatinine (p = 0.001), and decreased bicarbonate with either increased WBC count, increased creatinine, or decreased platelet count (p < 0.001).

Conclusions/Significance

Patients with fatal leptospirosis sought care earlier, but were not admitted for care any earlier than non-fatal patients. Combinations of routine laboratory values predictive of fatal



GOPEN ACCESS

Citation: Sharp TM, Rivera García B, Pérez-Padilla J, Galloway RL, Guerra M, Ryff KR, et al. (2016) Early Indicators of Fatal Leptospirosis during the 2010 Epidemic in Puerto Rico. PLoS Negl Trop Dis 10(2): e0004482. doi:10.1371/journal.pntd.0004482

Editor: Joseph M. Vinetz, University of California San Diego School of Medicine, UNITED STATES

Received: January 4, 2016

Accepted: February 1, 2016

Published: February 25, 2016

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative Commons CC0</u> public domain dedication.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This study was supported by the U.S. Centers for Disease Control and Prevention and Puerto Rico Department of Health. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

outcome should be considered in admission decision-making for patients with suspected leptospirosis.

Author Summary

Leptospirosis is a common tropical illness that results from exposure to the urine of animals infected with *Leptospira* bacteria. Because leptospirosis shares signs and symptoms with other common tropical illnesses such as dengue, identification of patients with leptospirosis can be challenging. Early identification of patients with leptospirosis is necessary to initiate antibiotic therapy and in some cases provide in-hospital management. During an epidemic of leptospirosis in Puerto Rico that occurred during a concomitant dengue epidemic, we identified leptospirosis patients by screening specimens from suspected dengue patients. Of 175 leptospirosis patients identified, 26 (15%) died. After comparing leptospirosis patients that died to patients of a similar age that were hospitalized but survived, we observed that fatal cases were more often sent home after their first medical visit. We next identified several routinely available laboratory values from patients' first medical visit that were associated with patients that died. Clinicians can use such laboratory values to diagnose and hospitalize leptospirosis patients at increased risk for fatal outcome.

Introduction

Leptospirosis is an emerging zoonosis caused by infection with bacterial spirochetes of the genus *Leptospira*, and is endemic throughout the tropics where >1 million cases and ~60,000 deaths occur annually [1, 2]. Human infection typically occurs through direct or indirect contact with the urine of infected animals [1]. Leptospirosis is typically a mild acute febrile illness (AFI); however, ~10% of patients progress to severe leptospirosis with acute kidney failure, jaundice, and/or pulmonary hemorrhage [1, 3]. The case-fatality rate for patients with severe leptospirosis ranges from 5-20% [4-6].

Due to similar clinical presentations, leptospirosis may be misdiagnosed as dengue [7-9]. Delayed or misdiagnosis of leptospirosis patients has been associated with increased mortality, potentially due to delayed administration of antibiotics [10-15]. Therefore, identification of early clinical markers of patients at risk for severe disease to thereby enable earlier patient admission may result in improved outcome. Severe thrombocytopenia, increased serum creatinine or BUN, hemoptysis, dyspnea, and jaundice have been associated with severe or fatal outcome in leptospirosis patients [5, 12, 14-18]; however, few studies have captured data from patients' entire clinical course to identify demographic characteristics, clinical findings, or missed opportunities in clinical management associated with poor outcome [12, 14]. Consequently, early clinical indicators of patients that have or will develop severe disease have not been well elucidated.

During 1990–2014, a total of 729 leptospirosis cases were reported to Puerto Rico Department of Health (PRDH), of which 78 (10.7%) were fatal (S1 Fig). Such surveillance enabled documentation of leptospirosis epidemics in 2006, 2007, and 2010. However, because of underreporting of leptospirosis [19], which is attributable in part to misdiagnosis as dengue [20-22], it is unclear if these data represent the true epidemiologic trends of leptospirosis. Factors associated with severe or fatal outcome in leptospirosis patients have not previously been investigated in Puerto Rico. To better understand the epidemiology of leptospirosis during the 2010 dengue epidemic in Puerto Rico [23], we conducted enhanced surveillance by performing leptospirosis diagnostic testing on specimens from suspected dengue patients. We also reviewed medical records from all health care visits of identified leptospirosis patients to identify demographic characteristics, clinical signs and symptoms, laboratory values, and clinical practices associated with fatal outcome.

Methods

Ethics statement

This study was approved by the Institutional Review Board at the Centers for Disease Control and Prevention (CDC) (protocol # 6285).

Data sources and diagnostic testing

Leptospirosis cases in Puerto Rico in 2010 were identified from four sources. First, suspected dengue cases reported via the Passive Dengue Surveillance System (PDSS) [24] that had no evidence of dengue virus (DENV) infection by rRT-PCR or anti-DENV IgM ELISA [23] (N = 2,519) were eligible to be tested for evidence of *Leptospira spp*. infection by microscopic agglutination test (MAT) [25] and polymerase chain reaction (PCR) with primers specific for Leptospira spp. LipL32 [26]. Specimens selected for leptospirosis testing (n = 1,133) came from cases for which either: a) paired acute and convalescent specimens were available (n = 654); or b) only a convalescent specimen was available and the case had reported fever, body pain or headache, and jaundice, hemorrhage, or pleural effusion (n = 479). Second, fatal leptospirosis cases were identified via the Enhanced Fatal AFI Surveillance System (EFASS) in which: a) serum or tissue specimens collected during autopsy were tested by MAT, PCR, or immunohistochemistry (IHC) [27]; and b) death certificates were reviewed for use of "leptospirosis" or "Weil's disease". Third, all leptospirosis cases reported to PRDH along with a positive diagnostic test result via the Notifiable Diseases Surveillance System (NDSS) were included. Last, two commercial laboratories were queried for leptospirosis cases that tested positive by IgM dot blot. Cases identified through more than one data source with matching names and dates of birth were considered a single case.

Definitions

A *laboratory-positive leptospirosis patient* was defined as a person that had evidence of infection with *Leptospira spp*. by detection of: i) antigen in a tissue specimen by IHC; ii) nucleic acid in a serum or tissue specimen by PCR; iii) \geq 4-fold rise in MAT titer in paired serum specimens; iv) MAT titer \geq 800 in a single serum specimen; v) anti-*Leptospira* IgM antibody at a private laboratory; or vi) MAT titer \geq 100 but <800 in a single serum specimen. A *confirmed leptospirosis patient* was defined by any of criteria i–v; a *probable leptospirosis patient* was defined by criteria vi. A *suspected fatal leptospirosis patient* was a person who died in Puerto Rico in 2010, had the word "leptospirosis" written on the death certificate, and had either: a) no leptospirosis diagnostic testing performed; or b) negative diagnostic testing performed at a commercial laboratory on a specimen collected within five days of illness onset.

Case-control study

Each fatal, laboratory-positive leptospirosis patient (i.e., cases) was matched by age within five years with up to three non-fatal, hospitalized, laboratory-positive leptospirosis patients (i.e., controls). All available medical records–including private office, out-patient clinic, emergency

department, and inpatient hospitalizations-during the episode of illness were reviewed. Controls that left the hospital against medical advice or had incomplete medical records were replaced.

Data analysis

The frequencies of clinical, demographic and laboratory data were calculated by performing descriptive analyses of all leptospirosis patients identified in 2010 and compared using Student's t-test or Chi squared test. Rates of leptospirosis by age group and municipality of residence were calculated using data from the 2010 United States Census [28]. Statistical differences and modeling of matched case-control data were performed using exact conditional logistic regression. Due to a limited number of matched pairs, several combinations of clinical lab results were considered for independent predictors of fatal outcome. Normal limits of laboratory values were defined by accepted standards [29].

All data analyses were conducted using SAS version 9.3 (SAS Institute Inc., Cary, NC), graphs were produced in SAS and Microsoft Excel (Microsoft Corp., Redmond, WA), and maps were created using ArcView (ESRI, Redlands, CA). Specimens were not anonymized prior to diagnostic testing to enable reporting of results to requesting physicians. Data included in the case-control study were anonymized prior to analysis.

Results

Identification of leptospirosis patients

Among 1,133 suspected but laboratory-negative dengue cases that were selected for leptospirosis diagnostic testing, 105 (9.3%) were laboratory-positive (<u>S1 Table</u>). Among 802 specimens from patients tested for leptospirosis at a private laboratory, 56 (7.0%) were positive. A total of 57 non-fatal leptospirosis patients were reported via NDSS in 2010, and laboratory diagnostic evidence was provided for 15 (26%). After consolidating individual patients identified by multiple systems, a total of 149 non-fatal, laboratory-positive leptospirosis patients were identified in Puerto Rico in 2010 (4.0 non-fatal patients per 100,000 residents), of which 91 (61%) were confirmed and 58 (39%) were probable leptospirosis patients. Dengue was ruled out for 134 (90%) non-fatal leptospirosis patients by rRT-PCR and/or IgM ELISA [23]; one apparent coinfection was identified in which DENV-1 was detected by RT-PCR and anti-*Leptospira spp*. IgM antibody was detected at a private laboratory.

A total of 26 fatal leptospirosis patients were identified (0.7 fatal patients per 100,000 residents), of which 21 were confirmed and five were suspected leptospirosis patients; only two (7.7%) had been reported to PRDH. Fifteen fatal, laboratory-positive leptospirosis patients had available kidney and liver tissue specimens, and *Leptospira* antigen was detected by IHC in all 15. Dengue was ruled out in 18 (86%) of the fatal, laboratory-positive leptospirosis patients and in two (40%) of the fatal, suspected leptospirosis patients. Two patients with fatal DENV/*Leptospira spp.* co-infection were identified [30]. Among all 26 fatal leptospirosis patients, the most common reported causes of death included respiratory, cardiac, or renal failure, and septic shock (S2 Table).

MAT-positive specimens (n = 130) from laboratory-positive leptospirosis patients showed strongest reactivity to serogroups including Icterohaemorrhagiae (57%), Australis (11%), Mini (5%), Bataviae (4%), Canicola (4%), Cynopteri (2%), Pyrogenes (2%), Pomona (1%), Djasiman (1%), and Autumnalis (1%); 12% had strongest reactivity against more than one serogroup. Of four PCR-positive serum specimens from one fatal and three non-fatal patients, multi-locus sequence typing [31] identified six of seven alleles suggestive of *L. interrogans* serovar



Fig 1. Date of illness onset of fatal (n = 26) and non-fatal (n = 147) leptospirosis patients identified in **Puerto Rico, 2010*.** *If date of illness onset was unavailable, date of first specimen collection was used instead. Two non-fatal patients had no available date of illness onset or date of specimen collection.

doi:10.1371/journal.pntd.0004482.g001

Icterohaemorrhagiae/Copenhageni in the specimen from the fatal patient; MLST was not successful for the other specimens.

Demographics and epidemiology

Leptospirosis patients had illness onset in all months of the year (Fig 1). Peak incidence of identified fatal and non-fatal leptospirosis patients occurred in October, in association with the rainy season. Most (79%) fatal and non-fatal laboratory-positive leptospirosis patients were male. Leptospirosis patients were identified in all age groups (Fig 2). Incidence was highest in individuals aged 40–69 years and lowest in individuals aged >80 years. Fatal patients were significantly older than non-fatal patients (mean of 50 vs. 41 years; p = 0.02). Confirmed and probable non-fatal leptospirosis patients were not significantly different by age (p = 0.34) or month of illness onset (p = 0.35); however, more confirmed than probable non-fatal patients were male (85% vs. 68%; p = 0.02). Most non-fatal (59%) and fatal (92%) leptospirosis cases were reported to have been hospitalized. Mortality by age group was highest in those aged 60– 69 years (1.8 per 100,000 residents).

Fatal and non-fatal leptospirosis cases resided in both urban and rural municipalities across Puerto Rico (Fig 3). In the 59 (76%) municipalities for which cases were detected, incidence was highest in Patillas in the rainy southeast–where enhanced dengue surveillance was conducted at a community health center in 2010 [23]–and in the mountainous, agricultural center of the island. Incidence was lowest in Cabo Rojo in the arid southwest.

Risk factors for fatal outcome

A case-control study was conducted in which data from medical records were compared between 21 laboratory-positive fatal cases and 52 age-matched, laboratory-positive, hospitalized but non-fatal leptospirosis controls. Cases and controls did not differ significantly by sex,





doi:10.1371/journal.pntd.0004482.g002

occupation, or animal or environmental exposure history, nor by reported co-morbidities or chronic medical conditions (<u>S3 Table</u>).

Fatal cases first sought medical care sooner after illness onset than non-fatal controls, and more often sought care at a private or out-patient clinic (<u>Table 1</u>). Although controls first sought medical care at a hospital more frequently than cases, cases were more often admitted



Fig 3. Rate of leptospirosis patients (N = 155*) and number of fatal patients (n = 26) by municipality of residence, Puerto Rico, 2010. Rates were calculated by dividing case numbers by municipality-specific populations, and grouped by quintile. *municipality of residence was unavailable for 18 non-fatal patients

doi:10.1371/journal.pntd.0004482.g003

or referred for admission at the first visit. Cases and controls did not differ by day post-illness onset (DPO) of hospitalization or duration of hospital stay. Cases were more often admitted to

Table 1. Characteristics of clinical management of fatal and non-fatal leptospirosis patients, Puerto Rico, 2010.

	Fatal cases N = 21	Non-fatal controls		
		N = 52		
	n (%)	n (%)	P-value	
Medications taken before first seeking care				
Acetaminophen	9 (42.9)	16 (30.8)	0.42	
Antibiotic	0 (0.0)	5 (9.6)	0.33	
Other*	7 (33.3)	11 (21.2)	0.38	
First medical visit				
DPO, median (range)	2.5 (0–10)	5 (2–14)	<0.01	
Facility type				
Private Clinic	2 (9.5)	2 (3.9)	0.62	
Out-patient clinic	8 (38.1)	2 (3.9)	Reference	
Hospital	11 (52.4)	47 (92.2)	<0.01	
Outcome				
Sent home	8 (38.1)	44 (84.6)	<0.01	
Admitted or referred for admission	13 (61.9)	8 (15.4)	Reference	
Hospitalization				
DPO admitted	4.5 (2–10)	6 (2–14)	0.31	
Duration of hospitalization in days, median (range)	4 (1–37)	8 (0–150)	0.68	
Admitted to intensive care unit	16 (76.2)	21 (40.4)	0.01	
Intubated	19 (90.5)	7 (13.5)	<0.01	
Hemodialysis initiated	7 (33.3)	4 (7.7)	0.02	
Received a blood product [†]	13 (61.9)	28 (53.8)	0.57	
Diagnosis				
"Leptospirosis" in differential at first medical visit	9 (42.9)	5 (9.6)	0.01	
"Dengue" or "viral syndrome" in differential at first medical visit	15 (71.4)	31 (59.6)	0.25	
"Leptospirosis" in discharge diagnosis at first medical visit	4 (19.0)	14 (26.9)	0.76	
"Dengue" or "viral syndrome" in discharge diagnosis at first medical visit	7 (33.3)	26 (50.0)	0.18	
"Leptospirosis" mentioned in any medical record	18 (85.7)	31 (59.6)	0.03	
DPO first mentioned, median (range)	5 (2–10)	7 (2–26)	0.09	
DPH first mentioned, median (range)	0.5 (0–20)	1 (0–15)	0.30	
"Dengue" mentioned in any medical record	15 (71.4)	52 (100)	<0.01	
DPO first mentioned, median (range)	3 (1–11)	5 (2–14)	0.23	
DPH first mentioned, median (range)	0 (0–4)	0 (0–2)	0.08	
Medications				
Penicillin-derivative antibiotic	19 (90.5)	38 (73.1)	0.62	
DPO given, median (range)	4.5 (3–10)	6.0 (3–13)	0.23	
DPH given, median (range)	1 (0–3)	1.0 (0–6)	0.73	
Corticosteroid	9 (42.9)	33 (63.5)	0.82	
DPO given, median (range)	4 (1–7)	6.0 (2–14)	0.20	
DPH given, median (range)	0 (0–2)	0 (0–4)	0.58	

*Other = aspirin, non-steroidal anti-inflammatory drug, statin, anti-depressant, anti-histamine, histamine H2R antagonist, nitrate, sulfonylurea [†]Cases: 8 (38%) received platelets, 5 (24%) received packed red blood cells, and 4 (19%) received fresh frozen plasma; Controls: 18 (35%) received platelets, 15 (29%) received packed red blood cells, 2 (4%) received fresh frozen plasma, and 1 (2%) received plasma. Abbreviations: DPO = day post-illness onset; DPH = day post-illness onset of hospitalization; N = Normal

doi:10.1371/journal.pntd.0004482.t001

the intensive care unit, intubated, and received hemodialysis (p \leq 0.02). Blood products were administered to more than half of cases and controls.

Cases more often had leptospirosis included in the differential diagnosis at first medical visit (p = 0.01), whereas controls more often had "dengue" ever mentioned in any medical record (p < 0.01). The timing with which "leptospirosis" and "dengue" were mentioned post-illness onset and post-hospitalization did not differ between cases and controls. Antibiotics were administered to >70% of cases and controls. Corticosteroids were administered to roughly half of cases and controls, most frequently on the day of admission. The frequency, clinical setting (e.g., out-patient clinic vs. hospital), and timing of administration of both antibiotics and corticosteroids did not significantly differ between cases and controls.

Cases presented to first medical visit with either fever or cough less often than controls (<u>Table 2</u>). Similarly, cases less often developed fever throughout hospitalization. Most cases developed jaundice, edema, leg pain, hemoptysis, and altered mental status, while fewer than half of controls had these findings. Developing cyanosis and having a seizure were also associated with fatal outcome.

DPO of first laboratory values did not differ significantly between cases and controls. As compared to controls, at first medical visit cases had significantly elevated white blood cell (WBC) count, proportion of neutrophils, BUN, creatinine, and total bilirubin, and decreased bicarbonate and albumin (Fig 4, S4 Table). For cases, these values were also more frequently outside of normal ranges. Throughout the clinical course, cases had significantly elevated WBC count, proportion of neutrophils, BUN, and creatinine, and decreased hematocrit, bicarbonate, albumin, prothrombin time (PT), and partial thromboplastin time (PTT).

Multivariable model of early indicators of fatal outcome

Because fever and cough were the only early clinical signs and symptoms that were associated with fatal outcome and may be spurious findings (see <u>Discussion</u>), only laboratory values were included as parameters in the model. BUN and PTT were removed from the model due to higher specificity of creatinine for kidney injury as opposed to dehydration and infrequency of the test being requested at initial patient presentation, respectively. Clinical laboratory values significantly associated with fatal outcome at first presentation as compared to controls included: decreased serum bicarbonate with elevated serum creatinine, elevated WBC count, or decreased platelet count; and elevated WBC count with elevated serum creatinine (<u>Table 3</u>).

Discussion

Enhanced surveillance demonstrated a high rate of morbidity and mortality due to leptospirosis in Puerto Rico in 2010 (4.7 and 0.7 cases per 100,000 residents, respectively). Comparable incidences have been observed in other regions of the Caribbean that have conducted enhanced surveillance [32–36], which also demonstrated highest burden in older male agricultural workers and the unemployed [2, 36]. Although the patients identified in Puerto Rico reflected the expected clinical characteristics of severe leptospirosis (i.e., pulmonary hemorrhage, acute kidney injury, and/or septic shock with multi-organ failure), under recognition and underreporting of leptospirosis cases was prominent, as one-third of patients were never diagnosed with leptospirosis and two-thirds were not reported to public health authorities. These findings together demonstrate that leptospirosis remains a neglected tropical disease in Puerto Rico.

Several missed opportunities for early clinical intervention were identified in this study. First, although fatal cases sought care earlier and were more often diagnosed with leptospirosis at first medical visit; however, fatal patients less often first sought care at a hospital, and were not admitted to the hospital any sooner than non-fatal patients. Thus, delayed hospital admission may have contributed to fatal outcome, as has been previously reported [12, 13]. However, we saw no evidence that this delay was associated with the timing of initiation of antibiotic therapy, which did not differ between cases and controls. Although prospective

Table 2. Signs and symptoms of fatal (n = 21) and non-fatal (n = 52) leptospirosis patients at first presentation and during hospitalization, Pu	ierto
Rico, 2010.	

Signs or symptom	At first presentation			During hospitalization			
	Fatal	Non-fatal	P-value	Fatal	Non-fatal	P-value	
	(N = 21)	(N = 52)		(N = 21)	(N = 52)		
	n (%)	n (%)		n (%)	n (%)		
Fever*	10 (47.6)	31 (59.6)	0.01	15 (71.4)	49 (94.2)	0.03	
Weakness/lethargy	9 (42.9)	27 (51.9)	0.58	19 (90.5)	46 (88.5)	0.29	
Headache	9 (42.9)	26 (50.0)	0.13	14 (66.7)	41 (78.8)	1.00	
Eye pain	2 (9.5)	4 (7.7)	0.58	5 (23.8)	9 (17.3)	0.34	
Conjunctival suffusion	0 (0.0)	8 (15.4)	0.07	3 (14.3)	12 (23.1)	0.67	
Conjunctival hemorrhage	0 (0.0)	2 (3.8)	1.00	2 (9.5)	4 (9.6)	0.58	
Icteric sclera	6 (28.6)	10 (19.2)	1.00	14 (66.7)	21 (40.4)	0.37	
Jaundice	6 (28.6)	11 (21.2)	0.65	17 (81.0)	22 (42.3)	0.02	
Cyanosis	2 (9.5)	0 (0.0)	0.08	6 (28.6)	1 (1.9)	<0.01	
Petechia	1 (4.8)	9 (17.3)	0.65	6 (28.6)	15 (28.8)	0.76	
Purpura	3 (14.3)	3 (5.8)	0.17	5 (23.8)	8 (15.4)	1.00	
Rash	1 (4.8)	9 (17.3)	1.00	3 (14.3)	16 (30.8)	0.14	
Dehydration	8 (38.1)	22 (42.3)	0.06	16 (76.2)	41 (78.8)	1.00	
Edema	5 (23.8)	10 (19.2)	0.70	16 (76.2)	16 (30.8)	<0.01	
Effusion	3 (14.3)	10 (19.2)	0.61	7 (33.3)	18 (34.6)	1.00	
Ascites	2 (9.5)	4 (7.7)	1.00	4 (19.0)	5 (9.6)	0.31	
Cardiac effusion	1 (4.8)	1 (1.9)	0.33	1 (4.8)	1 (1.9)	0.33	
Anorexia	6 (28.6)	19 (36.5)	0.58	13 (61.9)	32 (61.5)	0.48	
Vomiting	10 (47.6)	18 (34.6)	1.00	16 (76.2)	30 (57.7)	0.14	
Diarrhea	10 (47.6)	14 (26.9)	0.53	15 (71.4)	29 (55.8)	0.13	
Hepatomegaly	2 (9.5)	6 (11.5)	1.00	4 (19.0)	10 (19.2)	1.00	
Splenomegaly	1 (4.8)	5 (9.6)	1.00	1 (4.8)	8 (15.4)	0.66	
Muscle pain	15 (71.4)	32 (61.5)	1.00	19 (90.5)	49 (94.2)	1.00	
Leg pain	7 (33.3)	11 (21.2)	0.25	15 (71.4)	20 (38.5)	0.01	
Dyspnea	6 (28.6)	14 (26.9)	0.66	18 (85.7)	27 (51.9)	0.07	
Cough	2 (9.5)	15 (28.8)	0.01	8 (38.1)	30 (57.7)	0.17	
Altered mental status	6 (28.6)	7 (13.5)	0.45	18 (85.7)	12 (23.1)	<0.01	
Meningitis	0 (0.0)	0 (0.0)	†	0 (0.0)	1 (1.9)	†	
Encephalitis	2 (9.5)	0 (0.0)	0.50	2 (9.5)	1 (1.9)	0.17	
Seizure	2 (9.5)	1 (1.9)	1.00	5 (23.8)	1 (1.9)	0.03	
Epistaxis	1 (4.8)	1 (1.9)	1.00	5 (23.8)	3 (5.8)	0.10	
Hematemesis	1 (4.8)	2 (3.8)	1.00	5 (23.8)	7 (13.5)	0.67	
Hemoptysis	2 (9.5)	4 (7.7)	0.50	11 (52.4)	8 (15.4)	0.03	
Hematuria	6 (28.6)	21 (40.4)	0.27	14 (66.7)	36 (69.2)	0.77	
Melena	2 (9.5)	4 (7.7)	1.00	6 (28.6)	10 (19.2)	0.67	
Cerebral bleed	0 (0.0)	0 (0.0)	†	0 (0.0)	1 (1.9)	†	

*Subjective or objective

[†]Cannot be calculated due to missing information.

doi:10.1371/journal.pntd.0004482.t002





Fig 4. Box plots of selected laboratory values of fatal (n = 21) and non-fatal (n = 52) leptospirosis patients included in a case-control study, Puerto Rico, 2010. Medical records from all health care visits were abstracted, and first and worst laboratory values (<u>S4 Table</u>) were compared. The median value is indicated by the horizontal line within each box; mean value is indicated by the diamond; 25th and 75th interquartile range (IQR) are indicated by the bottom and top edges of the box, respectively; whiskers indicate the range of values within 1.5 times the value of the IQR. P values indicate statistical significant differences between fatal and non-fatal patients. Shaded horizontal lines indicate normal reference laboratory values.

doi:10.1371/journal.pntd.0004482.g004

clinical trials of antibiotics have not demonstrated a clear benefit to leptospirosis patient outcome [<u>37</u>], this should not preclude administration of antibiotics to patients with suspected

Model	Clinical test 1	OR (95% CI)	P-value	Clinical test 2	OR (95% CI)	P-value	Model P-value
1	Decreased serum bicarbonate	0.51 (0.12–0.91)	0.006	Elevated serum creatinine	2.93 (1.03–11.72)	0.042	<0.001
2	Decreased serum bicarbonate	0.50 (0.21–0.81)	<0.001	Decreased platelet count	1.02 (1.00–1.05)	0.015	<0.001
3	Decreased serum bicarbonate	0.19 (<0.01–0.78)	<0.001	Elevated WBC count	1.25 (1.03–1.75)	0.025	<0.001
4	Elevated WBC count	1.17 (1.03–1.38)	0.014	Elevated serum creatinine	2.09 (1.14–4.71)	0.010	0.001

Abbreviations: OR = odds ratio; CI = confidence interval; WBC = white blood cell

doi:10.1371/journal.pntd.0004482.t003

leptospirosis [10]. Last, roughly half of all leptospirosis patients were given corticosteroids, which may result in increased risk of hemorrhage and immunosuppression. A recent systematic review demonstrated no clear benefit to leptospirosis patient outcome by administering corticosteroids [38]; however, prospective clinical trials have yet to be conducted.

To improve recognition of leptospirosis and thereby mediate earlier admission for care, clinicians should be aware of patient characteristics and clinical indicators associated with severe leptospirosis. Most previous studies that identified risk factors associated with death due to leptospirosis relied on data collected during the final medical visit, which may be suboptimal for identification of early indicators of fatal outcome. After matching for age and status of hospitalization, no patient characteristics, including gender and history of smoking [15, 39], were significantly associated with fatal outcome in this study. Similar to previously studies [5, 8, 12, 16-18], we observed that jaundice, hemoptysis, acute kidney injury, and dyspnea or respiratory insufficiency were significantly associated with fatal outcome in this study, though not at initial medical visit. Therefore, the utility of these signs and symptoms may be limited in early identification of leptospirosis patients at risk for fatal outcome. Unexpected risk factors associated with fatal leptospirosis in this study were absence of cough and fever at first health care visit and lack of development of fever throughout hospitalization. Cough at initial presentation has been previously associated with protection from fatal outcome [12], though for unclear reasons. Potential explanations for lack of fever being associated with fatal outcome include incomplete capture of fever history, self-administration of antipyretics, or earlier entry into decompensated shock. Further studies should address the association of these signs and symptoms with fatal leptospirosis.

A prominent utility of this study was the association of common clinical laboratory values with fatal leptospirosis, specifically decreased bicarbonate with decreased platelet count and increased WBC count with elevated creatinine, all of which have been previously associated with severe leptospirosis [5, 8, 16-18, 40]. However, we did not observe that elevated serum potassium either at first presentation or at any point during hospitalization was associated with fatal outcome, as has previously been reported [40-43]. Nonetheless, the values of the laboratory markers of fatal outcome identified in this study tended to be farther outside of normal ranges at first presentation in fatal as compared to non-fatal patients, suggesting that patients with fatal leptospirosis may have progressed to severe disease more rapidly. In line with this, fatal patients were more likely to be diagnosed with leptospirosis earlier than were non-fatal patients, who were more likely to ever be diagnosed with dengue. Because previous studies associated elevated WBC count and elevated serum creatinine with leptospirosis as compared to dengue [20, 44-46], these clinical laboratory values may have utility in not only differentiating leptospirosis patients from dengue patients, but also in identifying leptospirosis patients at risk for poor outcome. Future studies should evaluate the prospective benefit of using such combinations of laboratory values to improve patient outcome through early identification and admission.

Compared to previous studies that have identified risk factors associated with severe or fatal outcome in leptospirosis patients, a major strength of this study was the design of the case-control study. By reviewing medical records from each health care visits made by patients included in the case-control study, and not solely those from the patients' hospitalization, we avoided biasing results towards points in patients' illness in which they were likely to be more clinically severe (i.e., at point of hospitalization). This also enabled identification of clinical indicators that would be of clinical utility before patients were hospitalized, which could thereby mediate more rapid diagnosis and/or hospitalization of patients at-risk for fatal outcome. Moreover, by closely matching patients by age we avoided identification of risk factors that may be associated with older populations. These aspects of study design together may account for some

differences in factors associated with fatal outcome identified by this study as compared to previous studies that did not control for age [3, 8, 12, 16, 17, 47]. Additional strengths of this study include: conducting surveillance for fatal leptospirosis cases by testing specimens collected during autopsy of patients that died following an AFI, without which many fatal cases would not have been diagnosed; and utilizing multiple surveillance systems to identify fatal and non-fatal leptospirosis patients and subsequently comparing them using a standardized instrument for chart abstraction.

Conversely, one limitation of this study is potential misclassification of some probable leptospirosis patients due to the presence of pre-existing neutralizing antibody. However, because several thousand suspected but dengue-negative cases reported to PRDH in 2010 were not tested for evidence of leptospirosis, the incidence of leptospirosis identified herein is likely an underestimate. Also, although previous studies have demonstrated that predictors of fatal leptospirosis include oliguria [8, 17, 18, 41, 48, 49] and anuria [12], we were unable to explore these factors due to the unavailability of routine clinical data on urine output. Moreover, due to limited sample size, we were also unable to identify specific cut-offs of clinical laboratory values associated with fatal outcome. Last, we were unable to evaluate DENV/*Leptospira spp*. co-infection as a risk factor for death since most leptospirosis cases were identified by screening suspected dengue cases that tested laboratory-negative for dengue.

Clinical trainings to improve early recognition of leptospirosis patients, interpretation of diagnostic test results, need for case reporting, and clinical management should be conducted among clinicians working in both out-patient and in-patient settings in Puerto Rico. Since improvements in case surveillance and clinical awareness have been associated with decreases in patient mortality due to leptospirosis [6], such trainings may also be needed in other areas of the tropics where clinical under recognition of leptospirosis may be high [2]. Population-based serosurveys should be conducted to accurately quantitate the burden of leptospirosis and identify modifiable risk factors associated with infection, including identification of the animal reservoirs that transmit *Leptospira spp*. to humans. Such findings can be used to develop educational campaigns to inform the public of population-specific strategies that can be employed to reduce their risk of leptospirosis.

Supporting Information

S1 Checklist. STROBE Checklist. (DOCX)

S1 Fig. Fatal (n = 78) and non-fatal leptospirosis (n = 651) patients reported to Puerto Rico Department of Health, 1990–2014. Year of illness onset or report were plotted for leptospirosis patients reported via the Notifiable Diseases Surveillance System. (TIF)

S1 Table. Method of leptospirosis diagnosis and rule out of dengue in fatal (n = 26) and non-fatal (n = 149) leptospirosis patients, Puerto Rico, 2010. (DOCX)

S2 Table. Reported causes of death^{*} in fatal leptospirosis patients, Puerto Rico, 2010. (N = 26). (DOCX)

S3 Table. Demographics, exposure, and medical history of matched fatal and non-fatal leptospirosis patients, Puerto Rico, 2010. (DOCX) S4 Table. Laboratory values and comparison results of matched fatal and non-fatal leptospirosis patients at presentation or worst recorded value during entire hospitalization, Puerto Rico, 2010.*

(DOCX)

Acknowledgments

We thank Drs. Mahesh Swaminathan, D. Fermín Argüello, George Han, and Julio Bracero for assistance with performing chart review, Laura Wright for assistance with map creation, and Nicole Pérez Rodríguez for assistance with data entry.

Author Contributions

Conceived and designed the experiments: TMS BRG MG DH SS KMT WAB. Performed the experiments: TMS BRG JPP RLG DH SR KMT WAB. Analyzed the data: TMS BRG KRR DH SR KMT WAB. Contributed reagents/materials/analysis tools: BRG RLG KRR DB. Wrote the paper: TMS BRG JPP RLG MG KRR DH SR SS DB KMT WAB.

References

- Ko Al, Goarant C, Picardeau M. Leptospira: the dawn of the molecular genetics era for an emerging zoonotic pathogen. Nature reviews Microbiology. 2009; 7(10):736–47. Epub 2009/09/17. doi: <u>10.1038/</u> <u>nrmicro2208</u> PMID: <u>19756012</u>.
- Costa F, Hagan JE, Calcagno J, Kane M, Torgerson P, Martinez-Silveira MS, et al. Global Morbidity and Mortality of Leptospirosis: A Systematic Review. PLoS neglected tropical diseases. 2015; 9(9): e0003898. doi: <u>10.1371/journal.pntd.0003898</u> PMID: <u>26379143</u>; PubMed Central PMCID: PMC4574773.
- Yersin C, Bovet P, Merien F, Clement J, Laille M, Van Ranst M, et al. Pulmonary haemorrhage as a predominant cause of death in leptospirosis in Seychelles. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2000; 94(1):71–6. Epub 2000/04/05. PMID: <u>10748905</u>.
- Panaphut T, Domrongkitchaiporn S, Vibhagool A, Thinkamrop B, Susaengrat W. Ceftriaxone compared with sodium penicillin g for treatment of severe leptospirosis. Clinical Infectious Diseases. 2003; 36 (12):1507–13. Epub 2003/06/13. doi: 10.1086/375226 PMID: 12802748.
- Dupont H, Dupont-Perdrizet D, Perie JL, Zehner-Hansen S, Jarrige B, Daijardin JB. Leptospirosis: prognostic factors associated with mortality. Clinical Infectious Diseases. 1997; 25(3):720–4. Epub 1997/10/06. PMID: <u>9314467</u>.
- 6. Edwards CN, Nicholson GD, Hassell TA, Everard CO, Callender J. Leptospirosis in Barbados. A clinical study. The West Indian medical journal. 1990; 39(1):27–34. Epub 1990/03/01. PMID: 2333695.
- Ellis T, Imrie A, Katz AR, Effler PV. Underrecognition of leptospirosis during a dengue fever outbreak in Hawaii, 2001–2002. Vector borne and zoonotic diseases. 2008; 8(4):541–7. Epub 2008/05/02. doi: <u>10.</u> <u>1089/vbz.2007.0241</u> PMID: <u>18447625</u>.
- Ko Al, Galvao Reis M, Ribeiro Dourado CM, Johnson WD Jr., Riley LW. Urban epidemic of severe leptospirosis in Brazil. Salvador Leptospirosis Study Group. Lancet. 1999; 354(9181):820–5. Epub 1999/ 09/15. PMID: <u>10485724</u>.
- Levett PN, Branch SL, Edwards CN. Detection of dengue infection in patients investigated for leptospirosis in Barbados. The American Journal of Tropical Medicine and Hygiene. 2000; 62(1):112–4. Epub 2000/04/13. PMID: <u>10761734</u>.
- Vinetz JM. A mountain out of a molehill: do we treat acute leptospirosis, and if so, with what? Clinical Infectious Diseases. 2003; 36(12):1514–5. Epub 2003/06/13. doi: <u>10.1086/375275</u> PMID: <u>12802749</u>.
- Bharti AR, Nally JE, Ricaldi JN, Matthias MA, Diaz MM, Lovett MA, et al. Leptospirosis: a zoonotic disease of global importance. Lancet Infectious Diseases. 2003; 3(12):757–71. Epub 2003/12/04. PMID: 14652202.
- Amilasan AS, Ujiie M, Suzuki M, Salva E, Belo MC, Koizumi N, et al. Outbreak of leptospirosis after flood, the Philippines, 2009. Emerging Infectious Diseases. 2012; 18(1):91–4. Epub 2012/01/20. doi: 10.3201/eid1801.101892 PMID: 22257492.
- 13. Goswami RP, Goswami RP, Basu A, Tripathi SK, Chakrabarti S, Chattopadhyay I. Predictors of mortality in leptospirosis: an observational study from two hospitals in Kolkata, eastern India. Transactions of

the Royal Society of Tropical Medicine and Hygiene. 2014; 108(12):791–6. doi: <u>10.1093/trstmh/tru144</u> PMID: <u>25359320</u>.

- Katz AR, Ansdell VE, Effler PV, Middleton CR, Sasaki DM. Assessment of the clinical presentation and treatment of 353 cases of laboratory-confirmed leptospirosis in Hawaii, 1974–1998. Clinical Infectious Diseases. 2001; 33(11):1834–41. Epub 2001/11/03. doi: <u>10.1086/324084</u> PMID: <u>11692294</u>.
- Tubiana S, Mikulski M, Becam J, Lacassin F, Lefevre P, Gourinat AC, et al. Risk factors and predictors of severe leptospirosis in new caledonia. PLoS Neglected Tropical Diseases. 2013; 7(1):e1991. Epub 2013/01/18. doi: <u>10.1371/journal.pntd.0001991</u> PMID: <u>23326614</u>; PubMed Central PMCID: PMC3542117.
- Reis EA, Hagan JE, Ribeiro GS, Teixeira-Carvalho A, Martins-Filho OA, Montgomery RR, et al. Cytokine response signatures in disease progression and development of severe clinical outcomes for leptospirosis. PLoS Neglected Tropical Diseases. 2013; 7(9):e2457. doi: <u>10.1371/journal.pntd.0002457</u> PMID: 24069500; PubMed Central PMCID: PMC3777885.
- Spichler AS, Vilaca PJ, Athanazio DA, Albuquerque JO, Buzzar M, Castro B, et al. Predictors of lethality in severe leptospirosis in urban Brazil. The American Journal of Tropical Medicine and Hygiene. 2008; 79(6):911–4. Epub 2008/12/05. PMID: 19052303; PubMed Central PMCID: PMC2640419.
- Hochedez P, Theodose R, Olive C, Bourhy P, Hurtrel G, Vignier N, et al. Factors Associated with Severe Leptospirosis, Martinique, 2010–2013. Emerging Infectious Diseases. 2015; 21(12):2221–4. doi: 10.3201/eid2112.141099 PMID: 26583702.
- Centers for Disease C, Prevention. Notes from the field: investigation of leptospirosis underreporting— Puerto Rico, 2010. MMWR Morbidity and Mortality Weekly Report. 2012; 61(22):421.
- Bruce MG, Sanders EJ, Leake JA, Zaidel O, Bragg SL, Aye T, et al. Leptospirosis among patients presenting with dengue-like illness in Puerto Rico. Acta Tropica. 2005; 96(1):36–46. Epub 2005/08/09. doi: 10.1016/j.actatropica.2005.07.001 PMID: 16083836.
- Sanders EJ, Rigau-Perez JG, Smits HL, Deseda CC, Vorndam VA, Aye T, et al. Increase of leptospirosis in dengue-negative patients after a hurricane in Puerto Rico in 1996 [correction of 1966]. The American Journal of Tropical Medicine and Hygiene. 1999; 61(3):399–404. Epub 1999/09/25. PMID: 10497979.
- Tomashek KM, Rivera A, Munoz-Jordan JL, Hunsperger E, Santiago L, Padro O, et al. Description of a large island-wide outbreak of dengue in Puerto Rico, 2007. The American Journal of Tropical Medicine and Hygiene. 2009; 81(3):467–74. PMID: <u>19706917</u>.
- Sharp TM, Hunsperger E, Santiago GA, Munoz-Jordan JL, Santiago LM, Rivera A, et al. Virus-specific differences in rates of disease during the 2010 Dengue epidemic in Puerto Rico. PLoS Neglected Tropical Diseases. 2013; 7(4):e2159. doi: <u>10.1371/journal.pntd.0002159</u> PMID: <u>23593526</u>; PubMed Central PMCID: PMC3617145.
- Noyd DH, Sharp TM. Recent Advances in Dengue: Relevance to Puerto Rico. Puerto Rico Health Sciences Journal. 2015; 34(2):65–70. PMID: 26061055.
- 25. Dikken H, Kmety E. Serological typing methods of leptospires. In: Bergan T, Norris J, editors. Methods in Microbiology. London: Academic Press; 1978. p. 259–307.
- Stoddard RA, Gee JE, Wilkins PP, McCaustland K, Hoffmaster AR. Detection of pathogenic Leptospira spp. through TaqMan polymerase chain reaction targeting the LipL32 gene. Diagnosis of Microbiological Infectious Diseases. 2009; 64(3):247–55. Epub 2009/04/28. doi: <u>10.1016/j.diagmicrobio.2009.03.</u> 014 PMID: 19395218.
- Zaki SR, Shieh WJ. Leptospirosis associated with outbreak of acute febrile illness and pulmonary haemorrhage, Nicaragua, 1995. The Epidemic Working Group at Ministry of Health in Nicaragua. Lancet. 1996; 347(9000):535–6. PMID: <u>8596276</u>.
- 28. United States Census Bureau. American FactFinder 2011 [cited 2012 9 January]. Available from: http://factfinder2.census.gov/faces/nav/jsf/pages/index.xhtml.
- Kratz A, Ferraro M, Sluss PM, Lewandrowski KB. Laboratory reference values. New England Journal of Medicine. 2004; 351(15):1548–63. doi: <u>10.1056/NEJMcpc049016</u> PMID: <u>15470219</u>.
- Perez Rodriguez NM, Galloway R, Blau DM, Traxler R, Bhatnagar J, Zaki SR, et al. Case Series of Fatal Leptospira spp./Dengue Virus Co-Infections-Puerto Rico, 2010–2012. The American Journal of Tropical Medicine and Hygiene. 2014. doi: 10.4269/ajtmh.14-0220 PMID: 25092820.
- Boonsilp S, Thaipadungpanit J, Amornchai P, Wuthiekanun V, Bailey MS, Holden MT, et al. A single multilocus sequence typing (MLST) scheme for seven pathogenic Leptospira species. PLoS Neglected Tropical Diseases. 2013; 7(1):e1954. doi: <u>10.1371/journal.pntd.0001954</u> PMID: <u>23359622</u>; PubMed Central PMCID: PMC3554523.

- Adesiyun AA, Baboolal S, Suepaul S, Dookeran S, Stewart-Johnson A. Human leptospirosis in the Caribbean, 1997–2005: characteristics and serotyping of clinical samples from 14 countries. Revista Panamericana de Salud Publica. 2011; 29(5):350–7. Epub 2011/06/29. PMID: <u>21709940</u>.
- Chadee DD, Mohan AR, Cumberbatch A, Adesiyun AA. Revised incidence of leptospirosis in Trinidad and Tobago, West Indies. Acta Tropica. 2010; 113(2):207. doi: <u>10.1016/j.actatropica.2009.12.004</u> PMID: <u>20102817</u>.
- Storck CH, Postic D, Lamaury I, Perez JM. Changes in epidemiology of leptospirosis in 2003–2004, a two El Nino Southern Oscillation period, Guadeloupe archipelago, French West Indies. Epidemiology and Infection. 2008; 136(10):1407–15. doi: <u>10.1017/S0950268807000052</u> PMID: <u>18096102</u>; PubMed Central PMCID: PMC2870739.
- Everard CO, Edwards CN, Everard JD, Carrington DG. A twelve-year study of leptospirosis on Barbados. European Journal of Epidemiology. 1995; 11(3):311–20. Epub 1995/06/01. PMID: 7493664.
- Everard CO, Bennett S, Edwards CN, Nicholson GD, Hassell TA, Carrington DG, et al. An investigation of some risk factors for severe leptospirosis on Barbados. The Journal of Tropical Medicine and Hygiene. 1992; 95(1):13–22. Epub 1992/02/01. PMID: 1740814.
- Brett-Major DM, Coldren R. Antibiotics for leptospirosis. The Cochrane database of systematic reviews. 2012; 2:CD008264. doi: 10.1002/14651858.CD008264.pub2 PMID: 22336839.
- Rodrigo C, Lakshitha de Silva N, Goonaratne R, Samarasekara K, Wijesinghe I, Parththipan B, et al. High dose corticosteroids in severe leptospirosis: a systematic review. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2014. doi: <u>10.1093/trstmh/tru148</u> PMID: <u>25266477</u>.
- Lopes AA, Costa E, Costa YA, Sacramento E, de Oliveira AR Junior, Lopes MB, et al. Comparative study of the in-hospital case-fatality rate of leptospirosis between pediatric and adult patients of different age groups. Revista del Instituto du Medicina Tropica du Sao Paulo. 2004; 46(1):19–24. Epub 2004/04/ 02. PMID: 15057329.
- 40. Marotto PC, Nascimento CM, Eluf-Neto J, Marotto MS, Andrade L, Sztajnbok J, et al. Acute lung injury in leptospirosis: clinical and laboratory features, outcome, and factors associated with mortality. Clinical Infectious Diseases. 1999; 29(6):1561–3. Epub 1999/12/10. doi: 10.1086/313501 PMID: 10585813.
- Panaphut T, Domrongkitchaiporn S, Thinkamrop B. Prognostic factors of death in leptospirosis: a prospective cohort study in Khon Kaen, Thailand. International Journal of Infectious Diseases. 2002; 6 (1):52–9. Epub 2002/06/05. PMID: <u>12044303</u>.
- Marotto PC, Ko AI, Murta-Nascimento C, Seguro AC, Prado RR, Barbosa MC, et al. Early identification of leptospirosis-associated pulmonary hemorrhage syndrome by use of a validated prediction model. The Journal of Infection. 2010; 60(3):218–23. Epub 2009/12/23. doi: <u>10.1016/j.jinf.2009.12.005</u> PMID: <u>20026189</u>.
- Esen S, Sunbul M, Leblebicioglu H, Eroglu C, Turan D. Impact of clinical and laboratory findings on prognosis in leptospirosis. Swiss Medical Weekly. 2004; 134(23–24):347–52. Epub 2004/08/20. doi: 2004/23/smw-10436. PMID: <u>15318284</u>.
- 44. Gregory CJ, Lorenzi OD, Colon L, Garcia AS, Santiago LM, Rivera RC, et al. Utility of the tourniquet test and the white blood cell count to differentiate dengue among acute febrile illnesses in the emergency room. PLoS Neglected Tropical Diseases. 2011; 5(12):e1400. doi: <u>10.1371/journal.pntd.</u> 0001400 PMID: <u>22163057</u>; PubMed Central PMCID: PMC3232191.
- LaRocque RC, Breiman RF, Ari MD, Morey RE, Janan FA, Hayes JM, et al. Leptospirosis during dengue outbreak, Bangladesh. Emerging Infectious Diseases. 2005; 11(5):766–9. Epub 2005/05/14. PMID: <u>15890136</u>.
- 46. Varma MD, Vengalil S, Vallabhajosyula S, Krishnakumar PC, Vidyasagar S. Leptospirosis and dengue fever: a predictive model for early differentiation based on clinical and biochemical parameters. Tropical Doctor. 2014; 44(2):100–2. doi: <u>10.1177/0049475513515212</u> PMID: <u>24311548</u>.
- Daher EF, Silva GB Jr., Karbage NN, Carvalho PC Jr., Kataoka RS, Silva EC, et al. Predictors of oliguric acute kidney injury in leptospirosis. A retrospective study on 196 consecutive patients. Nephron Clinical Practice. 2009; 112(1):c25–30. Epub 2009/04/04. doi: 10.1159/000210571 PMID: 19342866.
- Daher E, Zanetta DM, Cavalcante MB, Abdulkader RC. Risk factors for death and changing patterns in leptospirosis acute renal failure. The American Journal of Tropical Medicine and Hygiene. 1999; 61 (4):630–4. Epub 1999/11/05. PMID: <u>10548299</u>.
- Seguro AC, Lomar AV, Rocha AS. Acute renal failure of leptospirosis: nonoliguric and hypokalemic forms. Nephron. 1990; 55(2):146–51. Epub 1990/01/01. PMID: 2362627.